Author's response to reviews

Title: Severe hypercalcaemia and lymphoma in an HTLV-1 positive Jamaican woman: case report

Authors:

Veronica R Lyell (veronica.lyell@veronikon.co.uk)
Elham Khatamzas (elham4@hotmail.com)
Theresa J Allain (theresa.allain@googlemail.com)

Version: 2 Date: 26 April 2007

Author's response to reviews:

Missing author name inserted on title page (Elham Khatamzas)
'Written patient consent...' detailed on Acknowledgements section
figure 2 uploaded again as faulty first time